**Author's response to reviews**

**Title:** Randomised controlled trial examining the effect of exercise in people with rheumatoid arthritis taking anti-TNFalpha therapy medication

**Authors:**

Angela Reid (angelatreid@hotmail.com)
Audrey Brady (audbrady@yahoo.co.uk)
Catherine Blake (c.blake@ucd.ie)
Anne-Barbara Mongey (anne.b.mongey@ucd.ie)
Douglas J Veale (douglas.veale@ucd.ie)
Oliver FitzGerald (oliver.fitzgerald@ucd.ie)
Tara Cusack (t.cusack@ucd.ie)

**Version:** 11  **Date:** 12 January 2011

**Author's response to reviews:**

Small typographical changes made